Author's response to reviews

Title: Influenza vaccination of primary healthcare physicians is associated with vaccination coverages in their patients: a prevalence study

Authors:

Pere Godoy (pere.godoy@gencat.cat)
Jesús Castilla (jesus.castilla.catalan@cfnavarra.es)
José M Mayoral (jose.m.mayoral.sspa@juntadeandalucia.es)
Vicente Martín (vicente.martin@unileon.es)
Jenaro Astray (jenaro.astray@salud.madrid.org)
Núria Torner (nuria.torner@gencat.cat)
Diana Toledo (pmgripea@ub.edu)
Núria Soldevila (nuriasolde@hotmail.com)
Fernando González-Candela (fernando.gonzalez@uv.es)
Susana García-Gutierrez (SUSANA.GARCIA.GUTIERREZ@osakidetza.net)
José Diaz-Borrego (jose.diazborrego.sspa@juntadeandalucia.es)
Sonia Tamames (TamGomSo@jcyI.es)
Angela Domínguez (angela.dominguez@ub.edu)

Version: 2
Date: 27 November 2014

Author's response to reviews: see over
Dear Sir,
Please find enclosed the revised version of the manuscript entitled “Influenza vaccination of primary healthcare physicians is associated with vaccination coverages in their patients”. The manuscript has been revised taking into account all the comments.

1. Line Numbering: We have added the line numeration into the manuscript.

2. Trial Registration Number: Our study is not a Clinical Trial.

   As explained in the methods section the manuscript shows the results of an epidemiological observational study that was approved by the Ethics and Clinical Research Committee of the Jordi Gol Institute for Research in Primary Care. You can find the document attached into the “Confidential comments to the editors section”.

   Line 100: We have added the word “observational”.

   Line 253: We have removed the phrase “In addition, clinical trials show it is feasible to increase the vaccination of primary care physicians”.

   Line 255: We have added the word “observational”.

3 – 4. Availability of Supporting Data section and Deposition of data: We do not add supplementary information. If any reader is interested in our data we can facilitate.

Yours faithfully,

Dr. Pere Godoy

C/Castell de Farfanya 12
25198 Lleida,(Spain)

E-mail: pere.godoy@genca.cat